\documentclass{article}
%\usepackage[OT4,OT1,T1]{fontenc}
%\usepackage[latin1,cp1255]{inputenc}
\usepackage{array}
\makeatletter
%\usepackage{babel}
%\newcommand{\theauthor}[1]{\hfill\begin{otherlanguage}{english}{\it #1}\end{otherlanguage} }
%\font\scf=ptmr8r scaled 910
%\font\kukua=ptmr8r scaled 1370
%\font\kukub=ptmr8r scaled 1200
%\usepackage{ij}
%\usepackage{times}

\def\ {\thinspace}
\lineskiplimit=100pt 
\lineskip=6.9pt

\def\index\indexvolume#1\indexyear#2\endindextitle{
\rm\parindent=0pt%\overfullrule=0pt
\centerline{ISRAEL JOURNAL OF MATHEMATICS}
\vskip0.4cm
\centerline{TABLE OF CONTENTS, VOLUME #1\unskip, #2\unskip}
\vskip1.2cm
\frenchspacing}
\def\endindex{}


%\def\indexitem\indexauthor#1\indextitle#2\indexpage#3\endindexitem
%{\halign {\normalbaselines%
%\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%
%&\hbox to 1.3cm{\hfill ##}\cr%
%{\sc #1\unskip}\quad{\sl #2} & #3 \cr}\par}
%\def\indexauthor{}
%\def\endindexitem{}
%\def\indexpage{}
%\def\indextitle{}
\newcommand{\indexitem}[3]{\halign {\normalbaselines%
\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%
&\hbox to 1.3cm{\hfill ##}\cr%
{\sc #1\unskip}\quad{\sl #2} & #3 \cr}\par}

%\def\indexsee\indexauthor#1\partner#2\endindexitem
%{{\sc #1\unskip\enspace}{\sl See }{\sc #2\unskip}\par}

\newcommand{\indexsee}[2]{{\sc #1\unskip\enspace}{\sl See }{\sc #2\unskip}\par}

\makeatother

\begin{document}
\pagestyle{empty}

\index
\indexvolume 158
\indexyear 2007
\endindextitle

%3367
\indexitem{Abert, M. and L. Babai}{Finite groups of uniform logarithmic\break diameter}{193}

%3367
\indexsee{Babai, L.}{M. Abert}

%3374
\indexitem{Barvinok, A. and A. Samorodnitsky}{Random weighting, asymptotic counting, and inverse isoperimetry}{159}

%3395
\indexitem{Bayart, F. and L. Quarta}{Algebras in sets of queer functions}{285}

%3357
\indexitem{Begun, B. and A. del Junco}{Amenable groups, stationary measures and partitions with independent iterates}{41}

%3373
\indexitem{Bell, M. and W. Marciszewski}{On scattered Eberlein compact spaces}{217}
% 3398
\indexitem{Bentkus, V.}{On measure concentration for separately Lipschitz functions in product spaces}{1}

%3364
\indexitem{Buckley, S. M. and  D. A. Herron}{Uniform domains and capacity}{129}

%3382
\indexitem{Ershov, M.}{Finite presentability of $SL_1(D)$}{297}

%3370
\indexitem{Giambruno, A., D. La Mattina and V. M. Petrogradsky\break}{Matrix algebras of polynomial codimention growth}{367}

%3369
\indexitem{Haydon, R., A. Molt\'o and J. Orihuela}{Spaces of functions with countably many discontinuities}{19}

%3364
\indexsee{Herron, D. A.}{S. M. Buckley}

%3368
\indexitem{Hochman, M.}{Averaging sequences and abelian rank in amenable groups}{119}

%3366
\indexitem{Huang, W., X. D. Ye and G. Zhang}{Relative entropy tuples, relative u.p.e.~and c.p.e.~extensions}{249}

%3357
\indexsee{del Junco, A.}{B. Begun}

%335
\indexitem{Kerner, D.}{Erratum for: ``Enumeration of singular algebraic curves''}{379} 

%3370
\indexsee{La Mattina, D.}{A. Giambruno}

%3373
\indexsee{Marciszewski, W.}{M. Bell}

%3369
\indexsee{Molt\'o, A.}{R. Haydon}

%3381
\indexitem{Mostovoy, J. and J. M. P\'erez-Izquierdo}{Dimension filtration on loops}{105}

%3372
\indexitem{Nitzan-Hahamov, S. and A. Olevskii}{Sparse exponential systems: completeness with estimates}{205}

%3372
\indexsee{Olevskii, A.}{S. Nitzan-Hahamov}

%3369
\indexsee{Orihuela, J.}{R. Haydon}

%3380
\indexitem{Panaite, F. and M. D. Staic}{Generalized (anti) Yetter--Drinfeld modules as components of a braided T-category}{349}

%3381
\indexsee{P\'erez-Izquierdo, J. M.}{J. Mostovoy}

%3370
\indexsee{Petrogradsky, V. M.}{A. Giambruno}

%3450
\indexitem{Praeger, C. E. and C. Schneider}{Three types of inclusions of innately transitive permutation groups into wreath products in product action}{65}

%3395
\indexsee{Quarta, L.}{F. Bayart}

%3374
\indexsee{Samorodnitsky, A.}{A. Barvinok}

%3450
\indexsee{Schneider, C.}{C. E. Praeger}

%3371
\indexitem{Shapira, U.}{Measure theoretical entropy of covers}{225}

%3380
\indexsee{Staic, M. D.}{F. Panaite}

%3366
\indexsee{Ye, X. D.}{W. Huang}

%3366
\indexsee{Zhang, G.}{W. Huang}

\end{document}